Author's response to reviews

Title: Ganglioneuroma: A rare cause of adrenal incidentaloma: a case report

Authors:

mine adas (mineadas@yahoo.com)
bora koc (drborakoc@hotmail.com)
gokhan adas (gokhantolgaadas@gmail.com)
filiz ozulker (filizozulker@hotmail.com)

Version: 2 Date: 18 November 2013

Author's response to reviews: see over
Dear Editor;

I am enclosing here with a manuscript entitled “Ganglioneuroma; a rare cause of incidentaloma: a case report” submitted to “Journal of Medical case reports” for possible evaluation.

Ganglioneuroma is a rare cause of adrenal incidentaloma. In the literature there have been only few case reported. The different point of our case is SUV levels in PET scans. In the literature; it is known that SUV levels higher than 3 correspondence with the malignancy. However, our case SUV level is 4.1 correlated with the malignancy, the final pathology correlated the ganglioneuroma which is called as benign tumor. In this case, we pointed the attention to this situation.

With the submission of this manuscript I would like to undertake that the above mentioned manuscript has not been published elsewhere, accepted for publication elsewhere or under editorial review for publication elsewhere; and that my Institute’s Okmeydani Training and Research Hospital representative is fully aware of this submission.

Thank you for taking time to review my study, I look forward to a favourable reply.

Sincerely,

MD.Bora KOC
Okmeydani Training and Research Hospital
Department of General Surgery